Author’s response to reviews

Title: Girl child marriage, socioeconomic status, and undernutrition: Evidence from 35 countries in sub-Saharan Africa

Authors:

Yvette Efevbera (yvette@mail.harvard.edu)
Jacqueline Bhabha (jacqueline_bhabha@harvard.edu)
Paul Farmer (paul_farmer@hms.harvard.edu)
Günther Fink (guenther.fink@swisstph.ch)

Version: 2 Date: 10 Oct 2018

Author’s response to reviews:

Please see attached supplementary materials for letter with response.